Author's response to reviews

Title: Toe spreading ability in men with chronic pelvic pain syndrome

Authors:

Ugur Yilmaz (uyilmaz@u.washington.edu)
Ivan Rothman (nana@u.washington.edu)
Marcia A. Ciol (marciac@u.washington.edu)
Claire C. Yang (cyang@u.washington.edu)
Richard E. Berger (rberger@u.washington.edu)

Version: 5 Date: 6 June 2005

Author's response to reviews: see over
Editor,
BMC Urology

Re: MS: 1231936941395430
   Toe spreading ability in men with chronic pelvic pain syndrome

June 6, 2005

Dear Editor,

Thank you for accepting the manuscript for publication in BMC Urology.

In this revised version of our manuscript we made the necessary formatting changes requested. We used sentence case in the manuscript and removed all unnecessary capitals. We also wrote in detail the affiliations of each author. Concerning the table, we removed the highlighting and used bold-typing for emphasizing the cells.

I will be happy to answer any questions related to the manuscript.

Best regards,

Ugur Yilmaz, M.D.

University of Washington,
Department of Urology,
Box 356510
Seattle, WA 98195-6510